Author's response to reviews

Title: Factitious Lymphoedema - A psychiatric condition mimicking Reflex Sympathetic Dystrophy

Authors:

Nnamdi Nwaejike (justnnamdi@aol.com)
H A.P Archbold (poolerarchbold@aol.com)
Darrin Wilson (wilsondarrin@aol.com)

Version: 2 Date: 18 September 2007

Author’s response to reviews:

Dear Editor,

Thanks for your comments..

We have made the following revisions as advised:
- Change title ending so that it includes ': a case report'.
- Remove cover letter from the main manuscript.
- Change 'case report' subheading to 'case presentation'.
- Include a conclusion section to the manuscript.
- Include a Competing interests declaration
- Include an Authors contribution list
- Include an Acknowledgement list.
- Add a consent section. This should contain a statement to confirm that patient consent was received for the manuscript to be published.

We hope you find this satisfactory,

Sincerely

Nnamdi Nwaejike